Author's response to reviews

Title: No evidence of differential effects of SFA, MUFA or PUFA on post-ingestive satiety and energy intake: a randomised trial of fatty acid saturation

Authors:

Caroline M Strik (c.strik@auckland.ac.nz)
Fiona E Lithander (f.leahy@hotmail.com)
Anne-Thea McGill (at.mcgill@auckland.ac.nz)
Alastair K MacGibbon (a.macgibbon@fonterra.com)
Brian H McArdle (b.mcardle@auckland.ac.nz)
Sally D Poppitt (s.poppitt@auckland.ac.nz)

Version: 3 Date: 14 May 2010

Author's response to reviews: see over
Changes made to the PROOF copy_Strik et al_MS ID 1990030853260851

The author list has been revised reformatted into the required style, eg. Sally D Poppitt

Header has been removed from the ms

Underlining has been removed from the ms

Running title and key words have been removed from the ms

E-mail addresses have been reformatted and are shown below the author information on the title page

Highlighting has been removed from the ms

Abbreviations – have been formatted into a paragraph. Also abbreviations previously omitted have been added

Table 4 has been reformatted to Portrait

Figure 1 – cropped to minimise white space

Figure 2 – cropped to minimise white space

Figure 3 - cropped to minimise white space

Kind regards

[Signature]